Author's response to reviews

Title: Endoscopic resection of duodenal bulb neuroendocrine tumor larger than 10mm in diameter.

Authors:

Shozo Yokoyama (yokoyama@wakayama-med.ac.jp)
Katsunari Takifuji (takifuji@wakayama-med.ac.jp)
Masaji Tani (matani@wakayama-med.ac.jp)
Manabu Kawai (kawai@wakayama-med.ac.jp)
Teiji Naka (teiji@wakayama-med.ac.jp)
Kazuhisa Uchiyama (kuchi@wakayama-med.ac.jp)
Hiroki Yamaue (yamaue-h@wakayama-med.ac.jp)

Version: 4 Date: 9 June 2011

Author's response to reviews: see over
June 9, 2011

BioMed Central Editorial Team

MS: 7923410144762584
Dear BMC Gastroenterology Editorial;

Thank you for your mail of June 8, 2011. We changed the format of our manuscript to journal style as you suggested. Enclosed is a latest manuscript entitled, "Endoscopic resection of duodenal bulb neuroendocrine tumor larger than 10mm in diameter." by Yokoyama et al. for publication in *BMC Gastroenterology*. We really appreciate your help.

Sincerely yours,

Hiroki Yamaue, M.D.
Professor and Chairman,
Second Department of Surgery,
Wakayama Medical University, School of Medicine
811-1 Kimiidera, Wakayama 641-8510, Japan

E-mail: yamaue-h@wakayama-med.ac.jp
Fax: +81-73-446-6566, Tel: +81-73-441-0613